Author's response to reviews

Title: Pain and satisfaction during rigid cystoscopic ureteral stent removal: A prospective, randomized, pilot study

Authors:

Jae Heon Kim (piacekjh@hanmail.net)
Sun Young Park (sunnypark97@schmc.ac.kr)
Moon Gyu Kim (rlaansrb1@schmc.ac.kr)
Yun Seob Song (yssong@schmc.ac.kr)

Version: 2 Date: 20 March 2014

Author's response to reviews: see over
March 21, 2014

Dear Editor

We have recently pain and satisfaction during rigid cystoscopic ureteral stent removal according to several different analgesic methods. There have been many studies about pain during rigid or flexible cystoscopy, but there have been few studies about pain during cystoscopic stent removal. Although shorter in duration than cystoscopy, cystoscopic stent removal yields a similar pain to cystoscopy.

The title of our manuscript is “Pain and satisfaction during rigid cystoscopic ureteral stent removal: A prospective, randomized, pilot study.” We are sending an original manuscript and illustrations. The manuscript consists of 17 pages of text, 2 tables and 2 figures.

The paper has written according to the guidelines provided by **BMC urology**. The authors are consisted of 4 authors and they were all active participants of this manuscript. I would appreciate very much if you could review it for publication in **BMC urology**. The manuscript has never been published before nor it is under consideration for publication elsewhere. The manuscript has no connection to any companies or products.

Thank you for your kind consideration in advance and we are looking forward to hearing a positive reply from you.

According to you suggestions, we changed the author contributions, conflict of interest, and acknowledgment.

Sincerely yours,
Yun Seob Song, M.D., PhD.
Professor
Department of Urology, Soonchunhyang University Hospital
Seoul, Korea.